Author's response to reviews

Title: Implementation of a lifestyle intervention for type 2 diabetes prevention in Dutch primary care: opportunities for intervention delivery.

Authors:

Paulina WA Vermunt (prediabeteseindhoven@gmail.com)
Ivon EJ Milder (ivon.milder@rivm.nl)
F Wielaard (f.wielaard@chello.nl)
Caroline A Baan (caroline.baan@rivm.nl)
Jos DM Schelfhout (j.schelfhout@chello.nl)
Gert P Westert (g.westert@iq.umcn.nl)
Hans AM van Oers (hans.van.oers@rivm.nl)

Version: 2 Date: 20 February 2012

Author’s response to reviews:

Dear sir / madam,

As requested in your email from 18 February we included a statement of consent in the methods section of our manuscript. It can be found under ‘intervention protocol’ after the medical ethical approval number (M07-1705). We are sorry for the inconvenience,

yours sincerely,
Paulien Vermunt